Author's response to reviews

Title: A case-control study on the combined effects of p53 and p73 polymorphisms on head and neck cancer risk in an Italian population

Authors:

Paola Gallì (paola.galli@edu.rm.unicatt.it)
Gabriella Cadoni (gabriella.cadoni@rm.unicatt.it)
Mariangela Volante (mariangela.v@libero.it)
Emma De Feo (emmadefeo@yahoo.it)
Rosarita Amore (rosarita.amore@rm.unicatt.it)
Arianna Giorgio (ariannagiorgio@yahoo.it)
Dario Arzani (dario.arzani@rm.unicatt.it)
Gaetano Paludetti (gaetano.paludetti@rm.unicatt.it)
Gualtiero Ricciardi (wricciardi@rm.unicatt.it)
Stefania Boccia (sboccia@rm.unicatt.it)

Version: 7 Date: 15 April 2009

Author's response to reviews:

Dear Editor,

please find enclosed the revised version of the manuscript with some editorial changes made.

Many thanks for your assistance

Stefania Boccia